Author's response to reviews

Title: Islamic Fasting and Multiple Sclerosis

Authors:

Soodeh Razeghi Jahromi (razeghi@razi.tums.ac.ir)
Mohammad Ali Sahraian (msahrai@sina.tums.ac.ir)
Fereshteh Ashtari (f_ashtari@med.mui.ac.ir)
Hormoz Ayramloo (ayromlouh@gmail.com)
Massoud Etemadifar (etemadifar.m@gmail.com)
Majid Ghaffarpour (ghafarpour@sina.tums.ac.ir)
Ehsan Mohammadiannejad (Ehsan_neuro@yahoo.com)
Shahryar Nafissi (s_nafissi@yahoo.com)
Alireza Nickseresh (nikar7@yahoo.com)
Vahid Shaygannejad (shaygannejad@med.mui.ac.ir)
Mansoreh Togha (oghae@sina.tums.ac.ir)
Hamid Reza Torabi (drhamidtorabi@yahoo.com)
Shadi Ziaie (shadi_z73@yahoo.com)

Version: 4
Date: 15 March 2014

Author's response to reviews:

Dear Editor-in-Chief, BMC Neurology

I would like to submit our revised manuscript titled “Islamic Fasting and Multiple Sclerosis” for debate section of your journal.

The number of the authors is 13 and I have highlighted each author name and numbered in a vertical column.

Again thank you very much for giving us the opportunity to publish in BMC Neurology.

Yours sincerely,

Mohammad Ali Sahraian
Associate professor of neurology
Department of neurology, Sina Hospital,
Tehran University of Medical Sciences,
Hassan Abad Square, Tehran, Iran
Tel: 0098 912 313 4205
E.mail: msahrai@sina.tums.ac.ir